Author's response to reviews

Title: Epidemiologic Natural History and Clinical Management of Human Papillomavirus (HPV) Disease: A Critical and Systematic Review of the Literature in the Development of An HPV Dynamic Transmission Model

Authors:

Ralph P Insinga (ralph_insinga@merck.com)
Erik J Dasbach (erik_dasbach@merck.com)
Elamin H Elbasha (elamin_elbasha@merck.com)

Version: 5 Date: 22 July 2009

Author's response to reviews:

Dear BMC Editorial Office,

All major and minor revisions requested in your e-mail of 7/21/2009 have been incorporated into the uploaded version of the manuscript, excepting those minor revisions related to Figure Cropping and References.

Thank you,

Ralph Insinga